18.06 Hour Exam Marc 1996 Prof. S. Lee our Recitation Leader: our Recitation Time: our Name: Grading 1. 2. 3. 4. TOT AL: Do all our ork on these pages. No calculators or notes. oin alues (total of 100) are mark ed on the left margin. 1a. Are the ectors and linearly indep enden t? [4] Are these ectors erp endicular to eac other? Explain. 1b. Are the ectors 10 basis for Explain. [4] 1c. Are the ectors basis for the subspace describ ed the [8] 3-dimensional plane 0? Explain. 1d. Find the alue for for whic the ectors 12 10 do not [10] span 2. Consider the follo wing matrix, (a) Find the missing en tries of 21 31 32 [10] (b) Using the result in part(a), �nd the in erse of [10] 3. Let 10 (a) Giv an LU-factorization of [6] (b) Giv basis for the column space of [6] (c) Giv basis for the ro space of [6] (d) Giv basis for the ullspace of [6] (e) Giv the complete solution (general solution) to the equation Ax [6] in the form particular homogeneous where 14 4. The follo wing information is kno wn ab out and 18 In fact, Ax is alw ys some ultiple of (a) What is the dimension of the ullspace of [6] (b) What is the dimension of the ro space of [4] (c) What is the dimension of the left ullspace of [6] (d) Giv nonzero solution to Ax [8]